Author’s response to reviews

Title: The IASP Pain curriculum for undergraduate allied health professionals: educators defining competence level using Dublin Descriptors.

Authors:

Wim van Lankveld (wim.vanlankveld@han.nl)
Basema Afram (b.afram@han.nl)
Bart Staal (b.staal@han.nl)
Rob van der Sande (r.vandersande@han.nl)

Version: 1 Date: 28 Aug 2019

Author’s response to reviews:

In the letter send to me two points were raised.

1. A new author was added.
2. Declaration section was not complete.

As to point 1: we have not included a new author. I can not find anywhere in the system any indication that the system has received a new name. Most likely this is a technical error of the system as the new name (Dr.R1-Wim van Lankveld) strongly reembles my correct name (Dr. Wim van Lankveld).

as to point 2: Declaration section is not complete.

As suggested by the editors we have revised the Declarations section (lines 404-423). In the “ethical approval and consent to participate” section, it is now stated that informed consent in writing was obtained from all participants. Participants are experts participating in the study and they all gave consent to publish.

In the author's contribution it is now stated that all authors have read and approved the manuscript. The Declararion "Availability of data" is now renamed "Availability of data and materials".
We have now answered and given a response to each of the subheading according to the guidelines. Please indicate what went wrong exactly, or what is missing.

Regards,

Wim van Lankveld